



Over-the-Scope Clip Applications as First-Line Therapy in the Treatment of Upper Non-variceal Gastrointestinal Bleeding, Perforations, and Fistulas

Jiayu Qiu^{1†}, Jun Xu^{1†}, Yanxia Zhang^{1†}, Foqiang Liao¹, Zhenhua Zhu¹, Xu Shu¹, Youxiang Chen¹ and Xiaolin Pan^{1,2*}

¹ Department of Gastroenterology, The First Affiliated Hospital of Nanchang University, Nanchang, China, ² Human Genetic Resources Center, The First Affiliated Hospital of Nanchang University, Nanchang, China

OPEN ACCESS

Edited by:

Hemant Goyal, The Wright Center, United States

Reviewed by:

Shubhra Mishra, Post Graduate Institute of Medical Education and Research (PGIMER), India Bogdan Silviu Ungureanu, University of Medicine and Pharmacy of Craiova, Romania

> ***Correspondence:** Xiaolin Pan xlpan16@126.com

[†]These authors have contributed equally to this work and share first authorship

Specialty section:

This article was submitted to Gastroenterology, a section of the journal Frontiers in Medicine

Received: 05 August 2021 Accepted: 06 January 2022 Published: 15 February 2022

Citation:

Qiu J, Xu J, Zhang Y, Liao F, Zhu Z, Shu X, Chen Y and Pan X (2022) Over-the-Scope Clip Applications as First-Line Therapy in the Treatment of Upper Non-variceal Gastrointestinal Bleeding, Perforations, and Fistulas. Front. Med. 9:753956. doi: 10.3389/fmed.2022.753956 **Background:** The over-the-scope clip (OTSC) is an innovative device and has been successfully used in endoscopic treatment, however, there is a lack of clinical data from China. The aim of this study is to investigate the OTSC applications in the treatment of upper non-variceal gastrointestinal bleeding (UNVGIB), perforations, and fistulas in China.

Methods: In total, 80 patients were treated with one OTSC respectively as first-line therapy in our endoscopy center between January 2016 and November 2020. Among them, 41 patients had UNVGIB, 34 patients had perforations, and five patients had fistulas. The technical and clinical success rates were used to assess the efficacy of OTSC on the above diseases. In addition, we compared the hemostatic efficacy of OTSC with the standard endoscopic therapy in ulcer bleeding and Dieulafoy's lesion by propensity score matching analysis.

Results: In general, the OTSCs were applied successfully in all patients and achieved 100% (80/80) technical success. The clinical success of all patients was 91.3% (73/80). Among 41 patients with UNVGIB, the clinical success was 85.4% (35/41); 6 patients presented with recurrence. For patients of Dieulafoy's lesion and under antithrombotic therapy, we found that OTSC treatment had both efficient and reliable hemostasis effects. In addition, according to the characteristics of ulcers, site of bleeding lesion, and Blatchford score, all patients received similar and reliable clinical success rates. After propensity score matching, we found that OTSC treatment had low rebleeding rates when compared with standard endoscopic therapy in both Dieulafoy's lesion (15.0 vs 30.0%) and ulcer bleeding (17.6 vs 29.4%). Among 34 patients with perforations, the clinical success was 100% (34/34). Among five patients with fistulas, only one patient failed in maintaining the OTSC before esophageal fistula healing, and the clip achieved an overall clinical success of 80% (4/5).

Conclusion: The OTSC represents a safe and effective endoscopic therapy for UNVGIB, perforations, and fistulas as first-line treatment, especially for Dieulafoy's lesion or patients under antithrombotic therapy for UNVGIB, etc. However, OTSC application in these specific lesions or patients lacks adequate evidence as first-line treatment. Therefore,

1

further larger sample and multi-center clinical trials are required to improve its indications in clinical treatment.

Keywords: OTSC (over-the-scope clip), endoscopy, first-line therapy, upper non-variceal gastrointestinal bleeding (UNVGIB), perforation, fistula

INTRODUCTION

With the advancement of endoscopic techniques, many gastrointestinal lesions have been properly managed, but there are still challenges. Most upper non-variceal gastrointestinal bleeding (UNVGIB) could be treated by conventional endoscopic therapies, but recent studies have shown that some UNVGIB such as Dieulafoy's lesion and refractory bleeding could not be treated successfully and develop recurrence easily after standard endoscopic therapies such as epinephrine injections, hemoclips, or coagulation (1, 2). Equally, although many treatments have been attempted for the closure of perforations and fistulas, more and more iatrogenic gastric perforations are raised during endoscopic resection such as endoscopic mucosal resection (EMR) and endoscopic submucosal dissection (ESD), which could be handled by endoscopic closure and conservative management. However, there are still rare cases that need surgery (3, 4). Moreover, the closure of gastrointestinal fistulas is currently difficult because of the fibrosis tissue, inadequate opening width, and so on (5).

The over-the-scope clip (OTSC; OVESCO Endoscopy AG, Tuebingen, Germany), an innovative endoscopic full-thickness suturing device, has been developed and spread worldwide since it was firstly introduced for the closure of iatrogenic colon perforations in an animal experiment in 2007 (6). After that, in 2008, its effective application for UNVGIB and perforations in humans was confirmed in clinical experiences (7). In subsequent experimental studies, its indications have been further evaluated in many ways such as the closure of transgastric natural orifice transluminal endoscopic surgery (NOTES), perforations, fistulas, hemorrhage, and marking lesions before surgery in the gastrointestinal tract (8-14). For UNVGIB, the use of OTSC has emerged in recurrent ulcer bleeding treatment with promising results (15-17). However, data on use of OTSC as first-line therapy are very limited (17, 18). To date, the evidence on the efficacy of OTSC in Dieulafoy's lesion is derived from case series or small descriptive studies, meanwhile, studies on the comparison with standard endoscopic therapy are lacking (16, 18).

The OTSC was academically promoted officially in China and has become popular since 2014. However, the studies on OTSC application are mainly from European countries, and the published clinical data from China are still lacking (13). Here, we analyze and present the retrospective clinical study results of 80 patients, using the OTSC system for UNVGIB, perforations, and fistulas in a tertiary care hospital of China. Meanwhile, we investigate comparative outcomes of OTSC as first-line therapy versus standard endoscopic therapy in ulcer bleeding and Dieulafoy's lesion by using propensity score matching analysis.

MATERIALS AND METHODS

Patient Selection

We conducted a single-center retrospective study on the application of OTSC as first-line treatment in patients who were confirmed by endoscopy to have UNVGIB, perforations, and fistulas. A total of 80 patients underwent OTSC placement between January 2016 and November 2020 in our Endoscopy Center and each patient was treated with only one OTSC as first-line treatment respectively. To assess the efficiency of OTSC application as first-line treatment in UNVGIB (including large, fibrotic ulcer beds with obvious visible vessel or ulcers where the endoscopic treatment was difficult to perform, which may not be amenable to conventional endoscopic therapies, Dieulafoy's lesion, and other cause of UNVGIB), 41 patients were screened for eligibility. In addition, to further investigate the efficiency of OTSC application compared with the standard endoscopic therapy in ulcer bleeding and Dieulafoy's lesion of UNVGIB by propensity score matching analysis, 1,307 consecutive patients with UNVGIB who underwent endoscopic hemostasis were screened for eligibility at the same period. The inclusion criteria were as follows: (a) Patients with UNVGIB. (b) Patients treated with endoscopic hemostasis. The exclusion criteria were as follows: (a) Other cause of bleeding than ulcer bleeding and Dieulafoy's lesion. (b) This endoscopic therapy was not the initial treatment. (c) Patients with incomplete clinical information. (d) Patients with malignant lesions confirmed by pathology. (e) Endoscopic hemostasis with only epinephrine injection (endoscopic monotherapy of epinephrine injection is not recommended by recent guidelines) (15). Among them, they were divided into ulcer bleeding and Dieulafoy's lesion according to the lesion type. Each type was further divided into "OTSC" group and "standard endoscopic therapy" group according to whether OTSC was used or not. The flowchart of patient selection is shown in Figure 1. Meanwhile, to evaluate the efficiency of OTSC application as first-line treatment in perforations and fistulas, 34 patients with perforation and 5 patients with fistula were screened for eligibility. The inclusion criteria for perforations and fistulas were as follows: (a) Patients with perforations or fistulas. (b) Patients treated with OTSC as firstline treatment. The study was approved by the Ethics Committee of the First Affiliated Hospital of Nanchang University.

The informed consent of patients was obtained before all endoscopic treatment. The medical history of patients was collected and analyzed including their demographics, indications for OTSC interventions, the characteristic of lesions, recurrence of the original lesion, as well as the technical and clinical success. The definition of technical success is the successful application at the targeted lesion when using OTSC. Clinical success is known as the achievement of the intent outcome without the need



for additional therapies within 30 days during the follow-up. Recurrence was defined by the repeated symptoms of bleeding, perforation, or fistula after endoscopic treatment during the 30 days of follow-up (11, 19).

The OTSC System

The OTSC system consists of an over-the-scope clip, an applicator cap, a hand wheel, and a twin or anchor type grasper. The clip is mounted on the applicator cap and released by the hand wheel. It is made up of nitinol alloy which has the elasticity and memory that can restore its original shape after being released on the tissue. Then, the targeted lesion can be sutured to full-thickness by its teeth. Due to the different sizes of lesions, the OTSC has three sizes including 11, 12, and 14 mm. According to different indications, there are three different claw shapes such as blunt atraumatic type (a type), pointed traumatic edges (t type), and a special type for gastric wall closure (gc type). In this study, the 12 and 14 mm traumatic type (12/6t and 14/6t) were used in our institutions. Because the OTSC has the special occlusal structure, one advantage is the space between the teeth, which can prevent tissue necrosis by promoting blood supply. After installing the OTSC system and reaching the lesion, the twin grasper or anchor grasper is used to approximate the margins of the defect and pull the damaged tissue into the transparent applicator cap with the help of suction. When the targeted tissue is completely sucked into the cap, the hand wheel is turned to release the clip by tightening the thread. All the above procedures were performed by the experienced endoscopists who have completed formal training on the OTSC system in our Endoscopy Center.

Statistical Analysis

We divided the patients into two groups according to whether OTSC was used. All of the baseline characteristics were analyzed, as shown in Tables 3, 4. Propensity score matching was used to reduce selection bias of each group. Propensity score matching was estimated by using a multivariable logistic regression model with the following covariates: sex, age, lesion location, ulcer size, Forrest classification, antithrombotic therapy, hypertension, diabetes, and Blatchford score. All the categorical variables were reported as frequencies and percentages in the study and used to create a propensity score so as to match the "standard endoscopic therapy" group patients with the "OTSC" group. Patients were 1:1 matched using the nearest-neighbor method and with a caliper of 0.2. χ^2 tests or Fisher's exact tests were used for all of the categorical variables. P < 0.05 were considered significant. R statistical software version 4.1.0 (www.r-project.org) and SPSS version 23.0 (IBM; Chicago, IL, USA) were used for the statistical analyses.

RESULTS

We retrospectively analyzed the collected clinical data. During this period, 80 patients [52 men and 28 women with an average age of 54 years (range 18–88 years)] treated with OTSC were divided into three categories based on indications, which were UNVGIB (n = 41), perforations (n = 34), and fistulas (n = 5). The characteristics of the patients are presented in **Table 1**. All patients were treated with only one traumatic version OTSC (**Table 2**).

In general, the technical success was 80/80 (100.0%). The clinical success was 73/80 (91.3%), for six cases presented with recurrence of upper gastrointestinal bleeding and one patient failed in maintaining the OTSC before esophageal fistula healing. In addition, success was achieved in all perforated patients (**Table 2**).

Upper Non-variceal Gastrointestinal Bleeding

All 41 patients with UNVGIB were treated with one OTSC respectively. In this group, the technical success was 41/41 (100.0%). However, six patients experienced rebleeding after OTSC placement. Therefore, the clinical success rate was 35/41 (85.4%).

According to the type of bleeding, lesions were divided into three types: "Dieulafoy's lesion," "ulcer," and "wound bleeding after EMR" (**Table 2**). Both technical success and clinical success were 100% in "wound bleeding after EMR."

In the Dieulafoy's lesion group of 21 patients, the technical success was 21/21 (100%). Nevertheless, three patients had rebleeding from the original exposed vessel and the clinical success was 18/21 (85.7%) (**Figures 2A–D**). The rebleeding occurred from 1 to 5 days. One rebleeding patient who had

a Dieulafoy's lesion in the large diverticulum of descending duodenum underwent vascular interventional therapy after OTSC treatment, but the rebleeding was finally successfully stopped by tissue glue injection. One rebleeding patient accepted subsequent vascular interventional therapy, but the hemorrhage was eventually stopped by surgery. The last patient had a history of gastric cancer complicated with recurrence, refused any subsequent therapy, and eventually died.

For further evaluation of the hemostasis achieved using OTSC, we compared the OTSC treatment to standard endoscopic therapy in Dieulafoy's lesion (Table 3). In the unmatched cohort, 21 patients had OTSC placement, and 115 underwent standard endoscopic hemostasis. The OTSC group and the standard endoscopic therapy group differed with respect to antithrombotic therapy. Utilization of hemostasis with OTSC was more frequent in duodenum and antithrombotic therapy, whereas patients in the standard endoscopic therapy group were more frequently treated in the stomach. In order to mitigate the effects of baseline confounders, patients were matched into 20 pairs using propensity score matching. Covariates included in the model were sex, age, lesion location, antithrombotic therapy, comorbidities, and Blatchford score. In the matched cohort, rebleeding events were less common in the OTSC group (3/20, 15.0%) compared with the standard endoscopic therapy group (6/20, 30.0%), however, the rebleeding rates between the two groups were not significantly different (15.0 vs. 30.0%, P = 0.451).

There were 19 patients in the ulcer group, with 19/19 (100.0%) technical success. The clinical success of all 19 patients was 16/19 (84.2%). In terms of the characteristics of ulcers, Forrest Ia, Ib, and IIa each had one patient with recurrence of the original lesion, and the rebleeding time ranged from 1 to 4 days. The hemorrhages of three rebleeding patients were successfully stopped by endoscopic drug injection, interventional therapy, and surgical treatment, respectively. Depending on the site of bleeding lesion, the clinical success rate in the stomach was 11/13 (84.6%), in the duodenum was 18/21 (85.7%), and residual stomach after surgery was 6/7 (85.7%). In the assessment of the risk of rebleeding in all cases by Blatchford score, clinical success was 2/2 (100%) in two patients with low risk scores (<6 points), while 33/39 (84.6%) patients had moderate and high risk scores $(\geq 6 \text{ points})$. Depending on antithrombotic therapy, both groups had acceptable clinical success (used 85.7%, unused 85.3%).

For further evaluation of the hemostasis achieved using OTSC, we compared OTSC treatment to standard endoscopic therapy in ulcer bleeding (**Table 4**). In the unmatched cohort, 19 patients had OTSC placement, and 380 underwent standard endoscopic therapy. The OTSC group and standard endoscopic therapy group differed with respect to lesion location and Forrest classification. Utilization of hemostasis with OTSC was more frequent in the remnant stomach after surgery, Forrest Ia, and antithrombotic therapy, whereas patients in the standard endoscopic therapy group had more lesions in the duodenum with Forrest IIb classification. Patients were matched into 17 pairs using propensity score matching, and the covariates included in the model were sex, age, site of bleeding, size of ulcer, Forrest classification, antithrombotic therapy, comorbidities, and Blatchford score. In the matched cohort, rebleeding events

TABLE 1 Overview on the characteristics and success rates of patients with different indications.	
--	--

Group	Patients (n)	Sex (male/female)	Median age (years) (min-max)	Technical success	Recurrence (n)	Clinical success
Bleeding	41	35/6	55 (18–76)	41/41 (100%)	6	35/41 (85.4%)
Perforations	34	12/22	51 (19–88)	34/34 (100%)	0	34/34 (100%)
Fistulas	5	5/0	62 (55–71)	5/5 (100%)	1	4/5 (80.0%)
Total	80	52/28	54 (18–88)	80/80 (100%)	7	73/80 (91.3%)

TABLE 2 | Indications and recurrence of OTSC placement.

Indication	n	Type of clip (a/t/gc)	Technical success	Recurrence (n)	Clinical success
Bleeding	41	(0/41/0)	41/41 (100%)	6	35/41 (85.4%)
Туре					
Dieulafoy's lesion	21	(0/21/0)	21/21 (100%)	3	18/21 (85.7%)
Ulcer	19	(0/19/0)	19/19 (100%)	3	16/19 (84.2%)
Forrest la	5	(0/5/0)	5/5 (100%)	1	4/5 (80%)
Forrest Ib	6	(0/6/0)	6/6 (100%)	1	5/6 (83.3%)
Forrest IIa	8	(0/8/0)	8/8 (100%)	1	7/8 (87.5%)
Wound bleeding after EMR	1	(0/1/0)	1/1 (100%)	0	1/1 (100%)
Location					
Stomach	13	(0/13/0)	13/13 (100%)	2	11/13 (84.6%)
Duodenum	21	(0/21/0)	21/21 (100%)	3	18/21 (85.7%)
Remnant stomach after surgery	7	(0/7/0)	7/7 (100%)	1	6/7 (85.7%)
Blatchford score					
<6 (low risk)	2	(0/2/0)	2/2 (100%)	0	2/2 (100%)
\geq 6 (moderate and high risk)	39	(0/39/0)	39/39 (100%)	6	33/39 (84.6%)
Antithrombotic therapy					
Yes	7	(0/7/0)	7/7 (100%)	1	6/7 (85.7%)
No	34	(0/34/0)	34/34 (100%)	5	29/34 (85.3%)
Perforations	34	(0/34/0)	34/34 (100%)	0	34/34 (100%)
Duodenal ESD	12	(0/12/0)	12/12 (100%)	0	12/12 (100%)
Gastric ESD	20	(0/20/0)	20/20 (100%)	0	20/20 (100%)
lleocecal EMR	1	(0/1/0)	1/1 (100%)	0	1/1 (100%)
Duodenal perforation	1	(0/1/0)	1/1 (100%)	0	1/1 (100%)
Fistulas	5	(0/5/0)	5/5 (100%)	1	4/5 (80%)
Tracheoesophageal fistula	1	(0/1/0)	1/1 (100%)	0	1/1 (100%)
Gastrobrochial fistula	1	(0/1/0)	1/1 (100%)	0	1/1 (100%)
Esophageal fistula	2	(0/2/0)	2/2 (100%)	1	1/2 (50%)
Duodenal fistula	1	(0/1/0)	1/1 (100%)	0	1/1 (100%)

OTSC, over-the-scope clip; a, atraumatic; t, traumatic; gc, gastric wall closure type; EMR, endoscopic mucosal resection; ESD, endoscopic submucosal dissection.

were less common in the OTSC group (3/17, 17.6%) compared with the standard endoscopic therapy group (5/17, 29.4%), however, the rebleeding rates between the two groups were not significantly different (17.6 vs. 29.4%, P = 0.688).

Perforations

A total of 34 patients with gastrointestinal perforations were treated with OTSC successfully (**Figures 3A–D**). Among these patients, one case had an iatrogenic perforation opposite to the duodenal papilla during ERCP, and was sutured by OTSC successfully. The remaining 33 cases had iatrogenic perforations

caused by ESD and EMR (32 ESD and 1 EMR). Most of the removed tumors were stromal tumors, as well as ectopic pancreas and leiomyomas, which were all underwent fullthickness resection. Before using OTSCs, some perforations were first closed by titanium clips and nylon cords but failed. A total of 20 cases were located in the stomach, 13 cases in the duodenum, and one case in the ileocecum. In brief, the OTSCs used for perforations both achieved 100% in technical and clinical success, regardless of etiology and location. During follow-up, there were no complications, such as re-perforation and bleeding.



FIGURE 2 | The OTSC application in the treatment of UNVGIB caused by Dieulafoy's lesion. (A,B) Dieulafoy's lesion was successfully treated with OTSC; (C) a follow-up endoscopy at 7 days; (D) a follow-up endoscopy at 2 months after OTSC placement confirmed clinical success.

TABLE 3 C	Clinical characteristics of the patients with Dieulafoy's lesion in the unmatched and matched cohorts.
-------------	--

Variables	Unmatched cohort			Matched cohort		
	OTSC (n = 21)	Standard endoscopic therapy ($n = 115$)	p-Value	OTSC (n = 20)	Standard endoscopic therapy (n = 20)	p-Value
Sex (male), n (%)	18 (85.7)	98 (85.2)	1.000	18 (90.0)	15 (75.0)	0.407
Age (≥60), <i>n</i> (%)	12 (57.1)	52 (45.2)	0.349	11 (55.0)	10 (50.0)	1.000
Location, n (%)			0.267			0.227
Stomach	11 (52.4)	80 (69.6)		11 (55.0)	15 (75.0)	
Duodenum	7 (33.3)	22 (19.1)		6 (30.0)	5 (25.0)	
Remnant stomach after surgery	3 (14.3)	13 (11.3)		3 (15.0)	0 (0.0)	
Antithrombotic therapy, <i>n</i> (%)	5 (23.8)	6 (5.2)	0.014	4 (20.0)	4 (20.0)	1.000
Hypertension, n (%)	5 (23.8)	19 (16.5)	0.532	5 (25.0)	4 (20.0)	1.000
Diabetes, n (%)	1 (4.8)	5 (4.3)	1.000	1 (5.0)	0 (0.0)	1.000
Blatchford score \geq 6, <i>n</i> (%)	21 (100.0)	107 (93.0)	0.609	20 (100.0)	17 (85.0)	0.231
Rebleeding, n (%)	3 (14.3)	28 (24.3)	0.405	3 (15.0)	6 (30.0)	0.451
Clinical success	85.7%	75.7%		85.0%	70.0%	

OTSC, over-the-scope clip. Statistical analysis was performed with χ^2 tests or Fisher's exact tests. p < 0.05 were considered significant.

Fistulas

Five patients were treated with OTSCs due to fistulas, and each patient also used only one OTSC (**Figures 4A–C**). In this group of patients, the technical success was 5/5 (100%), and the clinical success was 4/5 (80%). The clinical failure patient had an esophageal fistula after thoracic surgery because of a foreign body in the esophagus. The fistula was successfully closed with one OTSC for the first treatment. However, the OTSC later failed when the patient had recurrent symptoms 1 month later. During the second treatment, a covered metallic stent was chosen to block the fistula which finally solved the issue. Another four patients had a tracheoesophageal fistula and gastrobrochial fistula after surgical repair of a duodenal bulb perforation, and an esophageal fistula caused by a foreign body in the esophagus, all of them were cured by OTSCs in one treatment.

DISCUSSION

To our knowledge, OTSC provides an effective and available alternative to manage various indications such as UNVGIB, perforations, and fistulas (20–25). However, clinical results were

mainly from Europe, and there is a lack of data from East Asia, especially in China. Here, we evaluated a single-center experience about the application of OTSC in a Chinese endoscopy unit and also obtained favorable results for the use of OTSC. The overall technical and clinical success rates in our study were 100.0% (80/80) and 91.3% (73/80), respectively.

We found that OTSCs have achieved efficient and reliable hemostasis in UNVGIB, including Dieulafoy's lesion, ulcers, and other causes. Dieulafoy's lesion is characterized by a large submucosal artery without an associated ulcer (26). Identifying Dieulafoy's lesions could be challenging for the intermittent nature of bleeding and the absence of surrounding mucosal abnormalities. At present, endoscopic interventions are the main and primary treatment approach (27). However, the use of conventional endoscopic methods may be sometimes challenging because of the location of the lesion, size, and high risk of rebleeding, which may require angiographic embolization or on rare occasions surgical intervention for definitive hemostasis (27). At present, several studies reported only small cases on the use of OTSC for Dieulafoy's lesion, and achieved a reliable clinical effect (28-32). Our study reported 21 patients with Dieulafoy's lesion treated by OTSC, as far as we know, which was the

Variables	Unmatched cohort			Matched cohort			
	OTSC (<i>n</i> = 19)	Standard endoscopic therapy ($n = 380$)	p-Value	OTSC (<i>n</i> = 17)	Standard endoscopic therapy ($n = 17$)	<i>p</i> -Value	
Sex (male), n (%)	16 (84.2)	317 (83.4)	1.000	14 (82.4)	13 (76.5)	1.000	
Age (≥60), <i>n</i> (%)	8 (42.1)	151 (39.7)	1.000	7 (41.2)	7 (41.2)	1.000	
Location, n (%)			0.001			0.109	
Stomach	2 (10.5)	80 (21.0)		1 (5.9)	5 (29.4)		
Duodenum	13 (68.4)	293 (77.1)		12 (70.6)	11 (64.7)		
Remnant stomach	4 (21.0)	7 (1.8)	0.157	4 (23.5)	1 (5.9)		
after surgery							
Size (≥1 cm), <i>n</i> (%)	11 (57.9)	155 (40.8)	0.157	11 (64.7)	9 (52.9)	0.728	
Forrest, n (%)			0.000			0.224	
la	5 (26.3)	23 (6.0)		4 (23.5)	3 (17.6)		
lb	6 (31.6)	265 (69.7)		6 (35.3)	9 (52.9)		
lla	8 (42.1)	71 (18.7)		7 (41.2)	3 (17.6)		
llb	0 (0.0)	21 (5.5)		0 (0.0)	2 (11.8)		
Antithrombotic therapy,n (%)	3 (15.8)	28 (7.4)	0.176	1 (5.9)	2 (11.8)	1.000	
Ulcer history, n (%)	5 (26.3)	72 (18.9)	0.384	5 (29.4)	6 (35.3)	1.000	
Hypertension, n (%)	4 (21.0)	89 (23.4)	1.000	4 (23.5)	5 (29.4)	1.000	
Diabetes, n (%)	4 (21.0)	34 (8.9)	0.095	4 (23.5)	3 (17.6)	1.000	
Blatchford score \geq 6, n (%)	18 (94.7)	348 (91.6)	1.000	16 (94.1)	16 (94.1)	1.000	
Rebleeding, n (%)	3 (15.8)	61 (16.0)	1.000	3 (17.6)	5 (29.4)	0.688	
Clinical success	84.2%	83.9%		82.4%	70.6%		

OTSC, over-the-scope clip. Statistical analysis was performed with χ^2 tests or Fisher's exact tests. p <0.05 were considered significant.





current maximum sample size in published studies. The results indicated that rebleeding occurred in three patients (14.3%, 3/21) during the 30 day follow-up after successful hemostasis, and achieved a clinical success rate of 85.7% (18/21). So far, most studies on the use of OTSC in Dieulafoy's lesion include a limited sample size and lack a control group (16). Therefore, we tried to overcome this limitation by using propensity score analysis to balance the confounding factors between the OTSC and standard endoscopic therapy groups in our study. The result showed that the rebleeding rate of the standard endoscopic therapy group was twice as high as in the OTSC group (30.0 vs. 15.0%), however, the difference was not statistically significant. It is possible that the

sample size was not large enough to be significant. To sum up, the OTSC could be raised as first-line treatment for Dieulafoy's lesion. Nevertheless, more large sample studies are required to prove its indications in clinical treatment.

At present, the studies found endoscopic treatment with OTSC to be superior to standard endoscopic therapy for patients with recurrent peptic ulcer bleeding, and was recommended in guidelines (15, 16, 33). However, the clinical data on OTSC treatment used as first-line therapy are limited to case series and retrospective studies (16, 33). Although there was one RCT evaluating the efficacy and safety of OTSC vs. standard endoscopic therapy for first-line treatment of UNVGIB recently



published by Jensen et al. (29). However, methodological limitations to this study must be noted, including the relatively limited samples, and the inclusion of Dieulafoy's lesions in addition to peptic ulcers. In our study, we tried to compare the hemostasis achieved for ulcer bleeding between OTSC and standard endoscopic therapy groups by using propensity score analysis to balance the confounding factors. The result showed that the rebleeding rate of the standard endoscopic therapy group was almost twice as high as in the OTSC group (29.4 vs. 17.6%), however, the difference was not statistically significant. Similar to Dieulafoy's lesions, it is possible that the sample size was not large enough to be significant. Based upon current studies, the OTSC is shown to have an advantage over standard endoscopic therapy as first-line therapy for ulcer bleeding. Meanwhile, more large sample studies, especially randomized controlled studies, are required to evaluate its indications in clinical treatment as well.

Previous studies have shown that the risk of rebleeding of ulcers is significantly increased in patients who underwent antithrombotic therapy and have a relatively low success rate of hemostasis under the conventional endoscopic treatment (34-36). Improved endoscopic treatment of UNVGIB in anticoagulated patients might be achieved by new devices such as the OTSC, which allow for better tissue apposition and compression of bleeding vessels. For the time being, data on OTSC use for UNVGIB in patients under antithrombotic therapy are currently limited, quite different, and lack Chinese data. At present, only two studies reported a rebleeding rate from 9.5 to 38.5% among patients under antithrombotic therapy after OTSC treatment, and the rebleeding was more frequent among those who received antithrombotic therapy. However, the differences were not statistically significant (37, 38). In our study we found that the rebleeding between patients under antithrombotic therapy were similar with the patients without antithrombotic therapy (14.3 vs. 14.7%), which was consistent with the above results. The above indicates that OTSC may have an advantage in patients under antithrombotic therapy and more research is needed concerning the use of OTSC in anticoagulated patients.

In our study, there were six patients suffered from rebleeding under OTSC treatment, and the duration was from 1 to 5 days. In the subgroup, the duration of rebleeding was 1-5 days for Dieulafoy's lesion and 1-4 days for ulcers, which seems to be no

difference. One clinical review reported that rebleeding typically occurs 1–4 days after initial conventional endoscopic therapy for Dieulafoy's lesion (39). For now, there are hardly any related studies that focus on the rebleeding time of Dieulafoy's lesion under OTSC treatment. However, whether OTSC treatment has an influence on the rebleeding time of Dieulafoy's lesion is not clear, and we will pay more attention to this in future.

With the development of endoscopic closure technology, it is rare that an iatrogenic perforation needs surgery. The effective treatment of OTSCs for iatrogenic perforations has been proposed in many previous studies (40-43). In general, the small iatrogenic perforation can be successfully closed by through-the-scope clips (TTSC) only (44). For large iatrogenic perforations, a nylon loop pouch suture has always been used. For this purpose, endoscopic closure of large procedure-related perforations using a single-channel endoscope was first proposed in our endoscopy center (45). In our study, the iatrogenic perforations were caused by EMR/ESD and ERCP, and were all closed by OTSC successfully. According to our follow-up, there was also no occurrence of delayed malignant events such as bleeding, perforation, and intestinal obstruction caused by the drop of the clip. Based on the 100% clinical success of iatrogenic perforation by OTSC and non-OTSC endoscopic therapy at the same period, we did not add counterparts for the treatment of OTSC in iatrogenic perforation.

Apart from gastrointestinal bleeding and perforations, OTSCs have been successfully used in fistulas, including tracheoesophageal, gastrobronchial, and esophageal fistulas (41, 46, 47). In our study, four fistulous patients were successfully treated with OTSCs while one developed recurrence during follow-up. The recurrent patient had an esophageal fistula after thoracic surgery because of a foreign body in the esophagus. Considering reasons for clinical failure, chronic fistulas lead to tissue necrosis and fibrosis around the fistula which makes it difficult for the clamp to bite, and insufficient nutritional supply after OTSC placement. Up to now, it is still difficult to manage GI fistulas with either endoscopic or surgical interventions and the optimal therapy is still being explored. Clinical success of the OTSC application in fistula management appears limited (14). Therefore, OTSC was rarely used in patients with fistulas in our endoscopic center and the sample size was only five. It is really difficult to conduct a comparative study based on so small a sample size, especially by using propensity score analysis, so we also did not add counterparts for the treatment of OTSC in patients with fistulas.

The OTSC is designed for full-thickness suture and is made of nickel-titanium alloy, which is the same material as cardiac and intracranial stents. Therefore, it can be worn for a long time and for life. During our follow-up in 80 patients, only one patient with a fistula experienced OTSC migration and resulted in symptom recurrence 1 month later. The OTSC was confirmed to be smoothly discharged from the body through X-ray, and there were no complications. Consistent with the previous reports, long-term OTSC attachment can be safe and effective, regardless of spontaneous detachment (21, 48).

However, our study still has the following limitations. On the one hand, the small number of related patients included, especially the closure of fistulas, may lead to a certain onesidedness in the study results. On the other hand, as this was a single-center retrospective data analysis, the clinical data we selected were all from one tertiary care center of China, which could not avoid the inherent regional selection bias and represent the situation of other countries and hospitals. Therefore, larger sample and multi-center clinical studies are needed.

CONCLUSIONS

In conclusion, our study confirms that the OTSC system plays a safe and effective role in UNVGIB, perforations, and fistulas as first-line treatment. However, for the clinical efficacy of OTSC treatment on some special lesions or patients, such as Dieulafoy's

REFERENCES

- Lai Y, Rong J, Zhu Z, Liao W, Li B, Zhu Y, et al. Risk factors for rebleeding after emergency endoscopic treatment of dieulafoy lesion. *Can J Gastroenterol Hepatol.* (2020) 2020:2385214. doi: 10.1155/2020/2385214
- Zhong C, Tan S, Ren Y, Lü M, Peng Y, Fu X, et al. Clinical outcomes of over-the-scope-clip system for the treatment of acute upper non-variceal gastrointestinal bleeding: a systematic review and meta-analysis. *BMC Gastroenterol.* (2019) 19:225. doi: 10.1186/s12876-019-1144-4
- Pissas D, Ypsilantis E, Papagrigoriadis S, Hayee B, Haji A. Endoscopic management of iatrogenic perforations during endoscopic mucosal resection (EMR) and endoscopic submucosal dissection (ESD) for colorectal polyps: a case series. *Therap Adv Gastroenterol.* (2015) 8:176–81. doi: 10.1177/1756283X15576844
- Kim GJ, Park SM, Kim JS, Ji JS, Kim BW, Choi H. Risk factors for additional surgery after iatrogenic perforations due to endoscopic submucosal dissection. *Gastroenterol Res Pract.* (2017) 2017:6353456. doi: 10.1155/2017/6353456
- Law R, Wong Kee Song LM, Irani S, Baron TH. Immediate technical and delayed clinical outcome of fistula closure using an over-the-scope clip device. *Surg Endosc.* (2015) 29:1781–6. doi: 10.1007/s00464-014-3860-8
- Schurr MO, Hartmann C, Ho CN, Fleisch C, Kirschniak A. An overthe-scope clip (OTSC) system for closure of iatrogenic colon perforations: results of an experimental survival study in pigs. *Endoscopy*. (2008) 40:584– 8. doi: 10.1055/s-2008-1077354
- Kirschniak A, Kratt T, Stuker D, Braun A, Schurr MO, Konigsrainer A, et al. new endoscopic over-the-scope clip system for treatment of lesions and bleeding in the GI tract: first clinical experiences. *Gastrointest Endosc.* (2007) 66:162–7. doi: 10.1016/j.gie.2007.01.034
- 8. Maekawa S, Nomura R, Murase T, Ann Y, Harada M. Complete closure of artificial gastric ulcer after endoscopic submucosal dissection by combined use

lesion or patients under antithrombotic therapy of UNVGIB, larger sample and multi-center experiences are eagerly needed.

DATA AVAILABILITY STATEMENT

The raw data supporting the conclusions of this article will be made available by the authors, without undue reservation.

ETHICS STATEMENT

The studies involving human participants were reviewed and approved by the Ethics Committee of the First Affiliated Hospital of Nanchang University. The Ethics Committee waived the requirement of written informed consent for participation. All included cases were recorded in the Human Genetic Resources Center of the First Affiliated Hospital of Nanchang University.

AUTHOR CONTRIBUTIONS

JQ collected data, analyzed relevant information, and drafted the manuscript. JX collected the data and performed the data analysis. YZ contributed to figures collection and manuscript revision. FL contributed to statistical analysis. XP, ZZ, XS, and YC clinically managed the patients. XP designed the study, critically revised the paper, and approved the final submission. All authors contributed to the article and approved the submitted version.

FUNDING

This work was supported by the Youth Science Foundation of Jiangxi Province (Grant No. 20192BAB215034).

of a single over-the-scope clip and through-the-scope clips (with videos). *Surg Endosc.* (2015) 29:500–4. doi: 10.1007/s00464-014-3725-1

- Donatsky AM, Andersen L, Nielsen OL, Holzknecht BJ, Vilmann P, Meisner S, et al. Pure natural orifice transluminal endoscopic surgery (NOTES) with ultrasonography-guided transgastric access and over-the-scope-clip closure: a porcine feasibility and survival study. *Surg Endosc.* (2012) 26:1952– 62. doi: 10.1007/s00464-011-2135-x
- Kirschniak A, Subotova N, Zieker D, Königsrainer A, Kratt T. The over-thescope clip (OTSC) for the treatment of gastrointestinal bleeding, perforations, and fistulas. Surg Endosc. (2011) 25:2901–5. doi: 10.1007/s00464-011-1640-2
- Wedi E, Gonzalez S, Menke D, Kruse E, Matthes K, Hochberger J. One hundred and one over-the-scope-clip applications for severe gastrointestinal bleeding, leaks and fistulas. *World J Gastroenterol.* (2016) 22:1844– 53. doi: 10.3748/wjg.v22.i5.1844
- Honegger C, Valli PV, Wiegand N, Bauerfeind P, Gubler C. Establishment of Over-The-Scope-Clips (OTSC(R)) in daily endoscopic routine. United European Gastroenterol J. (2017) 5:247–54. doi: 10.1177/20506406166 57273
- Kobara H, Mori H, Nishiyama N, Fujihara S, Okano K, Suzuki Y, et al. Overthe-scope clip system: a review of 1517 cases over 9 years. J Gastroenterol Hepatol. (2019) 34:22–30. doi: 10.1111/jgh.14402
- Bartell N, Bittner K, Kaul V, Kothari TH, Kothari S. Clinical efficacy of the over-the-scope clip device: a systematic review. *World J Gastroenterol.* (2020) 26:3495–516. doi: 10.3748/wjg.v26.i24.3495
- Laine L, Barkun AN, Saltzman JR, Martel M, Leontiadis GI. ACG clinical guideline: upper gastrointestinal and ulcer bleeding. *Am J Gastroenterol.* (2021) 116:899–917. doi: 10.14309/ajg.000000000001245
- Gralnek IM, Stanley AJ, Morris AJ, Camus M, Lau J, Lanas A, et al. Endoscopic diagnosis and management of nonvariceal upper gastrointestinal hemorrhage (NVUGIH): European Society of Gastrointestinal

Endoscopy (ESGE) Guideline - Update 2021. *Endoscopy.* (2021) 53:300–32. doi: 10.1055/a-1369-5274

- Buddam A, Rao S, Koppala J, Rangray R, Abdussalam A, Mukherjee S, et al. Over-the-scope clip as first-line therapy for ulcers with high-risk bleeding stigmata is efficient compared to standard endoscopic therapy. *Endosc Int Open*. (2021) 9:E1530–E5. doi: 10.1055/a-1526-0754
- Mangiafico S, Pigo F, Bertani H, Caruso A, Grande G, Sgamato C, et al. Overthe-scope clip vs epinephrine with clip for first-line hemostasis in non-variceal upper gastrointestinal bleeding: a propensity score match analysis. *Endosc Int Open*. (2020) 8:E50–E8. doi: 10.1055/a-1005-6401
- Chavan R, Nabi Z, Karayampudi A, Jagtap N, Asif S, Yaralagadda R, et al. Outcomes of over-the-scope clip application in various gastrointestinal indications: experience from a tertiary care in India. *Ann Gastroenterol.* (2020) 33:473–9. doi: 10.20524/aog.2020.0515
- Weiland T, Rohrer S, Schmidt A, Wedi E, Bauerfeind P, Caca K, et al. Efficacy of the OTSC System in the treatment of GI bleeding and wall defects: a PMCF meta-analysis. *Minim Invasive Ther Allied Technol.* (2020) 29:121–39. doi: 10.1080/13645706.2019.1590418
- Hu J, Yang Y, Ge N, Wang S, Guo J, Liu X, et al. Long-term assessment of overthe-scope clip (OTSC) behavior after gastric application. *Minim Invasive Ther Allied Technol.* (2020) 29:86–9. doi: 10.1080/13645706.2019.1590417
- 22. Wedi E, Fischer A, Hochberger J, Jung C, Orkut S, Richter-Schrag HJ. Multicenter evaluation of first-line endoscopic treatment with the OTSC in acute non-variceal upper gastrointestinal bleeding and comparison with the Rockall cohort: the FLETRock study. *Surg Endosc.* (2018) 32:307– 14. doi: 10.1007/s00464-017-5678-7
- Weiland T, Fehlker M, Gottwald T, Schurr MO. Performance of the OTSC System in the endoscopic closure of iatrogenic gastrointestinal perforations: a systematic review. Surg Endosc. (2013) 27:2258–74. doi: 10.1007/s00464-012-2754-x
- Zhong C, Tan S, Ren Y, Luo X, Xu J, Fu X, et al. Endoscopic management of iatrogenic gastrointestinal defects with the over-the-scope clip (OTSC) system: an updated systematic review. *Minim Invasive Ther Allied Technol.* (2021) 30:63–71. doi: 10.1080/13645706.2019.1683582
- Singhal S, Changela K, Papafragkakis H, Anand S, Krishnaiah M, Duddempudi S. Over the scope clip: technique and expanding clinical applications. J Clin Gastroenterol. (2013) 47:749–56. doi: 10.1097/MCG.0b013e318296ecb9
- Wang Y, Bansal P, Li S, Iqbal Z, Cheryala M, Abougergi MS. Dieulafoy's lesion of the upper GI tract: a comprehensive nationwide database analysis. *Gastrointest Endosc.* (2021). doi: 10.1016/j.gie.2020.12.015
- Nojkov B, Cappell MS. Gastrointestinal bleeding from Dieulafoy's lesion: clinical presentation, endoscopic findings, and endoscopic therapy. World J Gastrointest Endosc. (2015) 7:295–307. doi: 10.4253/wjge.v7.i4.295
- Manno M, Mangiafico S, Caruso A, Barbera C, Bertani H, Mirante VG, et al. First-line endoscopic treatment with OTSC in patients with high-risk nonvariceal upper gastrointestinal bleeding: preliminary experience in 40 cases. *Surg Endosc.* (2016) 30:2026–9. doi: 10.1007/s00464-015-4436-y
- Jensen DM, Kovacs T, Ghassemi KA, Kaneshiro M, Gornbein J. Randomized controlled trial of over-the-scope clip as initial treatment of severe nonvariceal upper gastrointestinal bleeding. *Clin Gastroenterol Hepatol.* (2021) 19:2315– 2323.e2. doi: 10.1016/j.cgh.2020.08.046
- Zeng Q, Dai JF, Cao H, Zhang S. Dieulafoy disease with gastric MALT lymphoma: a case report. *Medicine (Baltimore)*. (2020) 99:e22651. doi: 10.1097/MD.00000000022651
- Skinner M, Gutierrez JP, Neumann H, Wilcox CM, Burski C, Monkemuller K. Over-the-scope clip placement is effective rescue therapy for severe acute upper gastrointestinal bleeding. *Endosc Int Open.* (2014) 2:E37– 40. doi: 10.1055/s-0034-1365282
- 32. Soetikno R, Asokkumar R, Sim D, Sato T, Kaltenbach T. Use of the over-the-scope clip to treat massive bleeding at the transitional zone of the anal canal: a case series. *Gastrointest Endosc.* (2016) 84:168–72. doi: 10.1016/j.gie.2016.01.041
- Mullady DK, Wang AY, Waschke KA. AGA clinical practice update on endoscopic therapies for non-variceal upper gastrointestinal bleeding: expert review. *Gastroenterology*. (2020) 159:1120– 8. doi: 10.1053/j.gastro.2020.05.095
- Laine L, Jensen DM. Management of patients with ulcer bleeding. Am J Gastroenterol. (2012) 107:345–60; quiz 61. doi: 10.1038/ajg.2011.480

- Chang HY, Zhou M, Tang W, Alexander GC, Singh S. Risk of gastrointestinal bleeding associated with oral anticoagulants: population based retrospective cohort study. *BMJ.* (2015) 350:h1585. doi: 10.1136/bmj.h1585
- Huang ES, Strate LL, Ho WW, Lee SS, Chan AT. Long-term use of aspirin and the risk of gastrointestinal bleeding. *Am J Med.* (2011) 124:426– 33. doi: 10.1016/j.amjmed.2010.12.022
- Lamberts R, Koch A, Binner C, Zachäus M, Knigge I, Bernhardt M, et al. Use of over-the-scope clips (OTSC) for hemostasis in gastrointestinal bleeding in patients under antithrombotic therapy. *Endosc Int Open.* (2017) 05:E324– E30. doi: 10.1055/s-0043-104860
- Wedi E, von Renteln D, Gonzalez S, Tkachenko O, Jung C, Orkut S, et al. Use of the over-the-scope-clip (OTSC) in non-variceal upper gastrointestinal bleeding in patients with severe cardiovascular comorbidities: a retrospective study. *Endosc Int Open*. (2017) 05:E875–E82. doi: 10.1055/s-0043-105496
- Nguyen DC, Jackson CS. The Dieulafoy's Lesion: an update on evaluation, diagnosis, and management. J Clin Gastroenterol. (2015) 49:541–9. doi: 10.1097/MCG.00000000000321
- Khater S, Rahmi G, Perrod G, Samaha E, Benosman H, Abbes L, et al. Over-the-scope clip (OTSC) reduces surgery rate in the management of iatrogenic gastrointestinal perforations. *Endosc Int Open.* (2017) 5:E389– E94. doi: 10.1055/s-0043-104862
- Piyachaturawat P, Mekaroonkamol P, Rerknimitr R. Use of the over the scope clip to close perforations and fistulas. *Gastrointest Endosc Clin N Am.* (2020) 30:25–39. doi: 10.1016/j.giec.2019.08.002
- Mennigen R, Colombo-Benkmann M, Senninger N, Laukoetter M. Endoscopic closure of postoperative gastrointestinal leakages and fistulas with the Over-the-Scope Clip (OTSC). J Gastrointest Surg. (2013) 17:1058–65. doi: 10.1007/s11605-013-2156-y
- Nishiyama N, Mori H, Kobara H, Rafiq K, Fujihara S, Kobayashi M, et al. Efficacy and safety of over-the-scope clip: including complications after endoscopic submucosal dissection. World J Gastroenterol. (2013) 19:2752– 60. doi: 10.3748/wjg.v19.i18.2752
- Paspatis GA, Arvanitakis M, Dumonceau JM, Barthet M, Saunders B, Turino SY, et al. Diagnosis and management of iatrogenic endoscopic perforations: European Society of Gastrointestinal Endoscopy (ESGE) Position Statement -Update 2020. *Endoscopy*. (2020) 52:792–810. doi: 10.1055/a-1222-3191
- Zeng CY, Li GH, Zhu Y, Zhou XJ, Lv NH, Chen YX. Single-channel endoscopic closure of large endoscopy-related perforations. *Endoscopy*. (2015) 47:735– 8. doi: 10.1055/s-0034-1392043
- Dogan UB, Akin MS, Yalaki S. Endoscopic closure of tracheoesophageal fistulas with the over-the-scope clip system. *J Coll Physicians Surg Pak.* (2014) 24 (Suppl 3):S193–5. doi: 11.2014/jcpsp.s193s195
- Fischer A, Hoppner J, Utzolino S, Richter-Schrag HJ. Over-the-scope clip (OTSC) closure of a gastrobronchial fistula after esophagectomy. *Endoscopy*. (2014) 46 (Suppl 1) UCTN:E638–9. doi: 10.1055/s-0034-13 90729
- Goda Y, Kobara H, Kobayashi N, Nishiyama N, Fujihara S, Tada N, et al. Long-term outcomes of over-the-scope clip for refractory gastrointestinal diseases. *Minim Invasive Ther Allied Technol.* (2021) 11:1– 11. doi: 10.1080/13645706.2020.1867584

Conflict of Interest: The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's Note: All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Copyright © 2022 Qiu, Xu, Zhang, Liao, Zhu, Shu, Chen and Pan. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.